Author's response to reviews

Title: Odontogenic keratocyst with agenesis: a case report.

Authors:

Mariano Lacarbonara (mariano.lacarbonara@libero.it)
Giuseppe Marzo (giuseppe.marzo@cc.univaq.it)
Vitantonio Lacarbonara (antlac@tiscali.it)
Annalisa Monaco (annalisamonaco@yahoo.it)
Mario Capogreco (mario.capogreco@cc.univaq.it)

Version: 2 Date: 6 September 2013

Author's response to reviews: see over
Dear Editor-in-Chief,

I’m sending you our manuscript entitled “ODONTOGENIC KERATOCYST WITH AGENESIS: A CASE REPORT.” by Lacarbonara Mariano, Marzo Giuseppe, Lacarbonara Vitantonio, Monaco Annalisa e Capogreco Mario from the University of L’Aquila, Italy.

We believe is the first case report in the literature where a Caucasians patient, with hypodontia, manifested also a KCOT. This case report helps the reader to make a correct diagnosis, treatment and management of diseases with a multidisciplinary approach.

We would like to have the manuscript considered for publication in "Journal of Medical Case Reports", because your magazine is widely published and our research group has often used your magazine. We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with its submission to this journal. Please let me know of your decision at your earliest convenience.

With my best regards

Mariano Lacarbonara